Author's response to reviews

Title: Audio computer-assisted self-interviewing (ACASI) may avert socially desirable responses about infant feeding in the context of HIV

Authors:

Anthony K Waruru (awaruru@ke.cdc.gov)
Ruth Nduati (rduati@icconnect.co.ke)
Thorkild Tylleskar (thorkild.tylleskar@cih.uib.no)

Version: 7 Date: 15 July 2005

Author's response to reviews: see over
The Editor, Journal of Medical Informatics and Decision Making, Biomed Central,

Dear Editors,

Re: Manuscript for the Journal of Medical Informatics and Decision Making
Please find the reviewed manuscript by Anthony K. Waruru, Ruth Nduati and Thorkild Tylleskär entitled "Audio computer-assisted self-interviewing (ACASI) may avert socially desirable responses about infant feeding in the context of HIV”, that we are re-submitting for publication after review.

In response to Dr Elizabeth C. Moylan email requesting for additional formatting, the manuscript is being re-submitted with some editions.

Capitalization in page 2 for ‘Audio Computer-Assisted Self-Interviewing has been changed to ‘Audio computer-assisted self-interviewing (ACASI) and ‘Graphic User Interface’ to graphic user interface (GUI). Only acronyms have been abbreviated

Communication on this paper should be addressed to Anthony K. Waruru (see address below). We look forward to hearing from you.

Correspondence:

Anthony K. Waruru,                           Prof. Thorkild Tylleskär,
P.O. Box 1578,                                  Centre for International Health, University
040100, Kisumu, KENYA.                          of Bergen, Armauer Hansens Building,
Tel: +254 057 530117/8                          N-5021, Bergen, NORWAY.
Cell: +254 0722 841763                          Tel: +47 5597 4475
email: [awaruru@ke.cdc.gov]                     Fax: +47 5597 4979

email: [thorkild.tylleskar@cih.uib.no]